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Minutes for the November 2, 2011, CERES  

Production Oversight Board Meeting  
 

 

Meeting began at 1:00 p.m. in the main conference room at 1 Enterprise. 

Meeting Notes 

There were no comments or corrections on the minutes from the last meeting.   

Announcements   
NPP launched successfully! 

 

Open Action Items 

1. Discuss the ISCCP-D2like-FlxDay/Nit file names with Dave Doelling.  (2/16/11 – 

Jonathan) 

2. Provide answers to Raja’s questions she emailed recently.  (3/16/11 – Jonathan) 

 

FM5 Status 

CERES power-on scheduled for Tuesday, November 8. 

 

SCCR Discussion (SCCRs that have been submitted or updated since the last meeting) 

CERESlib 859 – Updated to include ERBE-like 868.. 

   

There were no issues identified, so this SCCR will be approved. 

 

Systems  

Web cluster is down.  It’s expected to come back up today, except for ceres.larc.nasa.gov. 

 

Dale asked about the ERBE-like Web files – how and when will they move from the test site to 

the real site?  No one knew the answer. 

   

Status of Open SCCRs  

859 – November 4
th

 delivery date. 

846 – November 4
th

 delivery date. 

801 – Cathy preparing for delivery.  She would like to deliver on November 4
th

, but she’s 

working on other things for Norman right now. 

804 – Tom will deliver the updated scripts by November 4
th

. 

763 – Need at least two more weeks to deliver.  Science data is still being validated. 

809 – Making some progress; still investigating problems.  Delivery will probably be next year. 

814 – Received Aqua scripts from Nelson today. 

867 – Incorporating standardized exit codes.  

865 – On hold in CM. 

870 – On hold in CM until Terra SSF processing is finished. 
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Additions to the Posted Reports  

 

Toolkit 

No updates to status report. 

 

CERESlib 

No updates to status report. 

   

Perl_Lib 

No updates to status report. 

 

CERES Processing System 

Iteration 1 – build data structures.   

 

Instrument 

1.4P3 – Deliver in 6 – 8 months.  Mark working on this. 

 

1.1P8 – Working on memory leak issue. 

 

FM4 opto-isolator having a problem; getting too hot. 

 

Still working on solar angle code. 

 

ValR23’s need to be validated. 

   

ERBE-like 

Working on Snow PGE on x86. 

 

Investigating “jump” that Norman is seeing that neither Nitchie or Dale are seeing. 

 

Clouds 

Supporting Pat and CREW-3. 

 

Working on Ed4. 

  

Convolution 

No updates to status report. 

    

Inversion 

No updates to status report. 

 

SARB 

Can’t read certain NetCDF files in ASDC_archive from SCF2.  Working with Nelson on this 

issue. 

 

SYNI ran in about half the time using Nelson’s job throttler. 
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TISA Gridding & GGEO 

No updates to status report. 

 

TISA Averaging 

No updates to status report. 

 

Production Processing Database/Automation 

No updates to status report. 

   

Ordering Tool 

No updates to status report. 

  

FM5 

No updates to status report. 

 

    

Meeting ended at 1:56 p. m. 


